

## Format for ANSWERING REVIEWERS



September 25, 2014

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 13113-edited.docx).

**Title:** "Second-Generation" Stem Cells For Cardiac Repair.

**Author:** Núñez García A, Sanz-Ruiz R, Fernández Santos ME, Fernández-Avilés F.

**Name of Journal:** *World Journal of Stem Cells*

**ESPS Manuscript NO:** 13113

1. Revision has been made according to the suggestions of the reviewer

- (1) Running title provided.
- (2) Author contributions provided
- (3) Telephone and fax numbers provided.
- (4) Format for reference citation amended.
- (5) Format of heading 2 amended.
- (6) Format for references in table 1 amended.
- (7) Format for text references amended.

Thank you again for publishing our manuscript in the *World Journal of Stem Cells*.

Sincerely yours,

A handwritten signature in black ink, appearing to read 'Ricardo Sanz-Ruiz', with a long horizontal stroke extending to the right.

Ricardo Sanz-Ruiz, MD  
Laboratorio de Hemodinámica  
Área de Cardiología Invasiva  
Servicio de Cardiología  
Hospital General Universitario Gregorio Marañón  
28007, Madrid, Spain.  
Fax: +34914265882  
E-mail: rsanzruiz@hotmail.com